\documentclass{template-jurnal} %Bagian ini diedit oleh editor Jurnal Matematika UNAND

\usepackage{multicol}
\newcommand{\ecp}{\vartriangleright_e}

\hyphenation{di-tulis-kan di-terang-kan}
%Perhatikan aturan penulisan dan ukuran huruf yang digunakan
\begin{document}
\begin{thebibliography}{0}
% Urutkan sesuai pustaka yang dikutip terlebih dahulu dalam tulisan

\bibitem{akh19} Akhter, S., Farooq, R., 2019, Metric dimension of fullerene graphs, \textit{Electron. J. Graph Theory Appl.}, Vol. \textbf{7}(1): 91--103 %graph

\bibitem{cha24} Chandran, S., Reji, T., 2024, Edge metric dimension of some Cartesian product of graphs, \textit{Proyecciones: J. Math}, Vol. \textbf{43}(3): 587--611 %var

\bibitem{cha00} Chartrand, G., Eroh, L., Johnson, M.A., Oellermann, O.R., 2000, Resolvability in graphs and the metric dimension of a graph, \textit{Discrete Appl. Math.}, Vol. \textbf{105}: 99--113 %fund

\bibitem{dud18} Dudenko, M., Oliynyk, B., 2018, On unicyclic graphs of metric dimension 2 with vertices of degree 4, \textit{Algebra Discrete Math.}, Vol. \textbf{26}(2): 256--269 %graph

%\bibitem{Goddard} W. Goddard, Static mastermind, \textit{J. Combin. Math. Combin. Comput.}, \textbf{47} (2003) 225--236.

\bibitem{har76} Harary, F., Melter, R., 1976, On the metric dimension of a graph, \textit{Ars Combin.}, Vol. \textbf{2}: 191--195 %fund

\bibitem{kuz21} Kuziak, D., Yero, I.G., 2021, Metric dimension related parameters in graphs: A survey on combinatorial, computational and applied results, \textit{arXiv:2107.04877v1} %surv

\bibitem{jav08} Javaid, I., Rahim, M.T., Ali, K., 2008, Families of regular graphs with constant metric dimension, \textit{Utilitas Mathematica}, Vol. \textbf{65}: 21--33 %cit

%\bibitem{Johnson1} M. Johnson, Structure-activity maps for visualizing the graph variables arising in drug design, \textit{J. Biopharm. Stat.}, \textbf{3} (1993) 203--236.

%\bibitem{Johnson2} M. Johnson, Browsable structure-activity datasets, \textit{Adv. in Mol. Similarity}, \textbf{2} (1998) 153--170.

\bibitem{kur23} Kurnia, R., Abrar, A.M., Syarifudin, A.G., Wijaya, V.R., Supu, N.A., Suwastika, E., 2023, On properties of prime ideal graphs of commutative rings, \textit{Barekeng: J. Math. Appl.}, Vol. \textbf{17}(3): 1463--1472 %graph

\bibitem{mar24} Maryati, T.K., Sobiruddin, D., Fatra, M., Hadiputra, F.F., 2024+, On metric dimension of edge comb product of vertex-transitive graphs, https://toc.ui.ac.ir/article\_28255.html, in press. %fund

\bibitem{red21} Redmond, S., Szabo, S., 2021, When metric and upper dimensions differ in zero divisor graphs of commutative rings, \textit{Discrete Math. Lett.}, Vol. \textbf{5}: 34--40 %graph

\bibitem{rin24} Rinurwati, Maharani, F.D., 2024, Bi-edge metric dimension of graphs, \textit{Int. J. Comput. Sci. Appl. Math.}, Vol. \textbf{10}(1): 38--40 %var

\bibitem{sla75} Slater, P., 1975, Leaves of trees, \textit{Proc. 6th Southeastern Conf. on Comb. Graph Theory Comput.}, Vol. \textbf{14}: 549--559 %fund

\bibitem{sla18} Slamin, Dafik, Waspodo, A.G., 2018, Distance domination number of graphs resulting from edge comb product, \textit{J. Phys.: Conf. Ser.}, Vol. \textbf{1022}: 012008

\bibitem{til21} Tillquist, R.C., Frongillo, R.M., Lladser, M.E., 2021, Getting the lay of the land in discrete space: A survey of metric dimension and its applications, \textit{arXiv:2104.07201v1} %surv

\bibitem{wij22} Wijaya, K., 2022, Dimensi metrik dari graf hasil identifikasi, \textit{J. Mat. UNAND}, Vol. \textbf{11}(3): 199--209 %graph

\bibitem{yul23} Yulianti, L., Welyyanti, D., Yanita, Fajri, M.R., Saputro, S.W., 2023, On the metric dimension of Buckminsterfullerene-net graph, \textit{Indones. J. Combin.}, Vol. \textbf{7}(2): 63--72 %graph

\end{thebibliography}


\markboth{T. K. Maryati et al.} %Jika lebih dari dua penulis, tuliskan sebagai Nama Penulis Pertama dkk.
{On Metric Dimension of Edge Comb Product of Symmetric Graphs}

%%%%%%%%%%%%%%%%%%%%% Publisher's Area please ignore %%%%%%%%%%%%%%%
%
\catchline{}{}{}{}{}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\title{On Metric Dimension of Edge Comb Product of Symmetric Graphs}

\author{T. K. Maryati$^{a}$\footnote{penulis korespondensi}, D. Sobiruddin$^{a}$, F. F. Hadiputra$^{b}$}

\address{$^{a}$ Department of
Mathematics Education, Syarif Hidayatullah State Islamic University Jakarta, Ciputat, Jakarta, Indonesia,\\
$^{b}$ School of Mathematics and Statistics, The University of Melbourne, Parkville, VIC 3010, Australia,\\
%$^{c}$ Instansi Penulis Ketiga.\\
email : \email{tita.khalis@uinjkt.ac.id, dindin.sobiruddin@uinjkt.ac.id, fhadiputra@student.unimelb.edu.au}}

% Jika instansi semua penulis sama, cukup tuliskan satu kali saja, tanpa menggunakan $^{a}$

\maketitle
\setcounter{page}{1} %bagian ini diedit oleh editor Jurnal Matematika UNAND

\begin{abstract}
\begin{center}
Diterima ..... \quad Direvisi ..... \quad Dipublikasikan ..... 
%tanggal-tanggal tersebut \textbf{dikosongkan} saja
\end{center}

\bigskip

%\textbf{Abstrak}. %Dalam bahasa Indonesia
%Abstrak tidak melebihi 250 kata. Tuliskan intisari dari makalah anda.

%\keywords{Tuliskan maksimal tiga buah kata kunci, tuliskan sesuai urutan abjad}
%\bigskip

\textbf{Abstract}. % Dalam bahasa Inggris
\textit{Consider a finite graph $G$ that is simple, undirected, and connected. Let $W$ be an ordered set of vertices with $|W| = k$. The representation of a vertex $v$ is defined as an ordered $k$-tuple that consists of the distances from vertex $v$ to each vertex in $W$. The set $W$ is called a resolving set for $G$ if the $k$-tuples for any two vertices in $G$ are distinct. The metric dimension of $G$, denoted by $dim(G)$, is the smallest possible size of such a set $W$. In this paper, we determine the metric dimension of edge comb product of trees with complete multipartites or petersen graphs.}

\noindent \textit{Keywords}: Edge Comb Product, Symmetric, Metric Dimension
\end{abstract}

\section{Introduction}
In a connected graph $G$, the distance $d(u, v)$ between any two vertices $u$ and $v$ is the length of the shortest path connecting them. Consider an ordered set of vertices $W = \{w_1, w_2, \ldots, w_k\}$ in $G$ and a vertex $v$ in $G$. The representation $r(v|W)$ of $v$ with respect to $W$ is given by the $k$-tuple $(d(v, w_1), d(v, w_2), \ldots, d(v, w_k))$. If each vertex in $G$ has a unique representation with respect to $W$, then $W$ is called a resolving set for $G$. The resolving set with the smallest size is called a basis for $G$, and its cardinality is referred to as the metric dimension of $G$, denoted by $\dim(G)$.

It is known that the concept of metric dimension was first introduced independently by Slater \cite{sla75} and by Harary and Melter \cite{har76}. They provided characterizations of the metric dimension for tree graphs, which were later proven using a different approach by Chartrand et al. \cite{cha00}. Studies have been conducted to consider the metric dimension of regular graphs \cite{jav08}, unicyclic graphs \cite{dud18}, fullerene graphs \cite{akh19}, zero divisor graphs \cite{red21}, identified graphs \cite{wij22}, and prime ideal graphs \cite{kur23}. Several applications of metric dimension can be seen in \cite{til21} and \cite{kuz21}.

In this paper, we determine the metric dimension of an edge comb product of symmetric graphs. In particular, we consider tree combined with either complete multipartites or petersen graph which are known to be symmetric.

\section{Known Results}
A complete multipartite graphs $K_{m \times n}$ is a graph which is obtained by taking $m$ set each containing $n$ vertices such that two vertices in the same set are not adjacent and two vertices in different sets are adjacent. Moreover, a generalized Petersen graph $P(n,m)$ is a graph with a vertex set
\begin{align*}
    V(P(n,m)) = \{u_i, v_i \ | \ i \in [1,n]\}
\end{align*}
and an edge set
\begin{align*}
    E(P(n,m)) = \{u_iu_{i+1}, v_iv_{i+m} \ | \ i \in [1,n]\}
\end{align*}
where the indices are taken modulo $n$. The metric dimension of generalized Petersen graph $P(n,2)$ has been determined by Javaid et al. \cite{jav08}.
\begin{theorem}\cite{jav08}\label{pet}
Let $P(n,2)$ be the generalized Petersen graph. Then $dim(P(n,2)) = 3$ for $n \ge 5$.
\end{theorem}

An \textit{automorphism} is a permutation of the vertices in $V(G)$ that preserves adjacency. A graph is \textit{vertex-transitive} if, for any two distinct vertices $v_1$ and $v_2$ in $V(G)$, there exists an automorphism that maps $v_1$ to $v_2$. Similarly, a graph is \textit{edge-transitive} if, for any two distinct edges $e_1$ and $e_2$ in $E(G)$, there exists an automorphism that maps $e_1$ to $e_2$. A graph is called \textit{symmetric} if it is both vertex-transitive and edge-transitive. In particular, Petersen graph $P(5,2)$ is well known for its symmetry, which poses as an interesting graph.

The edge comb product of two graphs $G_1$ and $G_2$, denoted by $G_1 \ecp G_2$, is constructed by taking one copy of $G_1$ and $|E(G_1)|$ copies of $G_2$, and attaching the $i$-th copy of $G_2$ to the $i$-th edge of $G_1$ \cite{sla18}. Most of the time, the choice of an edge in the product might alter the result of the graph. However, if the graph is symmetric, then the choice of the edge does not matter. Recently, the metric dimension of an edge comb product of vertex-transitive graphs was determined \cite{mar24}. One of the results is written below.

\begin{theorem}\cite{mar24}\label{low_bound_tree}
Let $T$ be a tree graph whose size is $m$, $H$ be a vertex-transitive graph, and $p$ be the cardinality of pendants in $T$. For any edge $e \in E(H)$, we have
\begin{align*}
    dim(T \ecp H) \ge m(dim(H)-2) + p.
\end{align*}    
\end{theorem}

\section{Main Results}
First, we will consider an edge comb product of tree graph with complete multipartites $K_{m \times n}$ for some positive integers $m$ and $n$. %Since $K_{2 \times 2} \cong C_4$, we only consider the case where $m + n \ge 5$.

\begin{theorem}
Let $m \ge 2$ and $n \ge 2$ such that $K_{m \times n} \not\cong C_4$. Let $T$ be a tree graph with size $h \ge 2$ and $p$ be the number of pendants in $G$. For any edge $e \in E(K_{m \times n})$, we have
\begin{align*}
    dim(T \ecp K_{m \times n}) = h(m(n-1)-2) + p.
\end{align*}
\end{theorem}
\begin{proof}
Let $K_{m \times n}^e$ be a complete multipartite graph that is attached to the edge $e$. This graph has the vertex set 
\begin{align*}
    V(K_{m \times n}^e) = \{v_{i,j}^e \ | \ i \in [1,m], j \in [1,n]\}
\end{align*} and the edge set 
\begin{align*}
    E(K_{m \times n}^e) = \{v_{i,j}^ev_{i',j'}^e \ | \ i,i' \in [1,m], i \ne i', j,j' \in [1,n]\}.
\end{align*}
Since the metric dimension of $K_{m \times n}$ is $m(n-1)$, then by Theorem \ref{low_bound_tree}, we have
\begin{align*}
    dim(T \ecp K_{m \times n}) \ge h(m(n-1)-2) + p.
\end{align*}
To prove $dim(T \ecp K_{m \times n}) \le h(m(n-1)-2) + p$, we need to set a proper vertex set that can be uniquely determined by most of other vertices. Without loss of generality, for edges $e \in E(T)$ containing pendant let $v_{1,2}^e \in K_{m \times n}^e$ be the vertex identified to other $K_{m \times n}^{e'}$ for some $e' \in E(T)$. Likewise, for edges $e \in E(T)$ which does not contain pendant let $v_{1,2}^e, v_{2,2}^e \in K_{m \times n}^e$ be the vertices identified to other $K_{m \times n}^{e'}$ for some $e' \in E(T)$. Let $Q$ be the vertex set that has $v_{1,2}^e$ for every pendant edges $e$ and $v_{1,2}^e, v_{2,2}^e$ for every non-pendant edges. Next, let 
\begin{align*}
    S = Q \bigcup_{\substack{e \in E(T) \\ i \in [1,m]}} v_{i,1}.
\end{align*}

Set $W = V(G \ecp K_{m \times n}) \setminus S$. It is obvious that vertices in $W$ has a unique representation. Hence, we can only consider vertices in $S$. Let $x$ and $y$ be vertices in $S$. We split the cases as follows.

\iffalse
%e be a pendant edge
\noindent \textit{Case 1.} Let $e$ be a pendant edge and $n = 2$. Consequently, $m \ge 3$.\\
\textit{Subcase 1.1.} Let $x = v_{i,k}^e$ and $y = v_{j,1}^e$ for $i < j$ and $k \in \{1,2\}$. We have %If $j \ne 2$ or $e$ is a pendant edge, then
\begin{align*}
    d(v_{i,k}^e,v_{j,2}^e) = 1 \ne 2 = d(v_{j,1}^e,v_{j,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(v_{j,1}^e \ | \ W).
\end{align*}
\textit{Subcase 1.2.} Let $x = v_{i,k}^e$ and $y \in \{v_{j,1}^{e'},v_{l,2}^{e'}\}$ with $e$ is a pendant edge, $e$ and $e'$ are adjacent such that $v_{i',2}^{e'} = v_{1,2}^e$ for some $i',k,l \in \{1,2\}$, and $j \ge 2$. If $i \le m-1$ then
\begin{align*}
    d(v_{i,k}^e,v_{m,2}^e) = 1 \ne 2 = d(y,v_{m,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}
Else, let $i = m$. This implies
\begin{align*}
    d(v_{i,k}^e,v_{2,2}^e) = 1 \ne 2 = d(y,v_{2,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}
\textit{Subcase 1.3.} Let $x = v_{i,k}^e$ and $y \in \{v_{j,1}^{e'},v_{l,2}^{e'}\}$ with $e$ and $e'$ are not adjacent, $k,l \in \{1,2\}$ and $j \ge 2$. We have
\begin{align*}
    d(v_{i,k}^e,v_{m,2}^e) \le 2 \ne 3 \le d(y,v_{m,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}
\fi

%e be a pendant edge or n \ge 3
\noindent \textit{Case 1.} Let $n \ge 3$ or $e$ be a pendant edge. In particular, if $n = 2$, then $m \ge 3$.\\
\textit{Subcase 1.1.} Let $x = v_{i,k}^e$ and $y = v_{j,l}^e$ for $i < j$ and $k,l \in \{1,2\}$. We have
\begin{align*}
    d(v_{i,k}^e,v_{j,n}^e) = 1 \ne 2 = d(v_{j,l}^e,v_{j,n}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(v_{j,l}^e \ | \ W).
\end{align*}
\textit{Subcase 1.2.} Let $x = v_{i,k}^e$ and $y \in \{v_{j,1}^{e'},v_{l,2}^{e'}\}$ with $e$ and $e'$ are adjacent such that $v_{i',2}^{e'} = v_{1,2}^e$ for some $i',k,l \in \{1,2\}$ and $j \ge 2$. If $i \le m-1$ then
\begin{align*}
    d(v_{i,k}^e,v_{m,n}^e) = 1 \ne 2 = d(y,v_{m,n}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}
Else, let $i = m$. This implies
\begin{align*}
    d(v_{i,k}^e,v_{m-1,n}^e) = 1 \ne 2 = d(y,v_{m-1,n}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}
\textit{Subcase 1.3.} Let $x = v_{i,k}^e$ and $y \in \{v_{j,1}^{e'},v_{l,2}^{e'}\}$ with $e$ and $e'$ are not adjacent, $k,l \in \{1,2\}$ and $j \ge 2$. This implies
\begin{align*}
    d(v_{i,k}^e,v_{m,2}^e) \le 2 \ne 3 \le d(y,v_{m,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}

%e be a non-pendant edge and n = 2
\noindent \textit{Case 2.} Let $e$ be a non-pendant edge and $n = 2$. Consequently, $m \ge 3$.\\
\textit{Subcase 2.1.} Let $x = v_{i,k}^e$ and $y = v_{j,l}^e$ for $i < j$ and $k,l \in \{1,2\}$. If $j \ge 3$, then 
\begin{align*}
    d(v_{i,k}^e,v_{j,2}^e) = 1 \ne 2 = d(v_{j,l}^e,v_{j,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(v_{j,1}^e \ | \ W).
\end{align*}
Else, let $j = 2$ which implies $i = 1$. There exists an edge $e^* \in E(T)$ such that $v_{i',2}^{e^*} = v_{1,2}^e$ and $v_{m,2}^{e^*} \in W$. If $k = 1$ then,
\begin{align*}
    d(v_{1,1}^e,v_{m,2}^{e^*}) = 1 \ne 2 = d(v_{j,l}^e,v_{m,2}^{e^*}) \implies r(v_{1,1}^e \ | \ W) \ne r(v_{j,l}^e \ | \ W).
\end{align*}
Otherwise, $k = 2$ implying
\begin{align*}
    d(v_{1,2}^e,v_{m,2}^{e^*}) = 3 \ne 2 = d(v_{j,l}^e,v_{m,2}^{e^*}) \implies r(v_{1,2}^e \ | \ W) \ne r(v_{j,l}^e \ | \ W).
\end{align*}
\textit{Subcase 2.2.} Let $x = v_{i,k}^e$ and $y \in \{v_{j,1}^{e'},v_{l,2}^{e'}\}$ with $e$ and $e'$ are adjacent such that $v_{i',2}^{e'} = v_{1,2}^e$ for some $i',k,l \in \{1,2\}$ and $j \ge 2$. If $i \le m-1$ then
\begin{align*}
    d(v_{i,k}^e,v_{m,2}^e) = 1 \ne 2 = d(y,v_{m,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}
Else, let $i = m$. There exists an edge $e^* \in E(T)$ such that $v_{i^*,2}^{e^*} = v_{1,2}^e$ for $i^* \in \{1,2\}$ and $v_{m,2}^{e^*} \in W$. Since $T$ is a tree, then $V(H^{e^*}) \cap V(H^{e'}) = \varnothing$. Therefore,
\begin{align*}
    d(v_{i,k}^e,v_{m,2}^{e^*}) = 2 \ne 3 = d(y,v_{m,2}^{e^*}) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}
\textit{Subcase 2.3.} Let $x = v_{i,k}^e$ and $y \in \{v_{j,1}^{e'},v_{l,2}^{e'}\}$ with $e$ and $e'$ are not adjacent, $k,l \in \{1,2\}$ and $j \ge 2$. This implies
\begin{align*}
    d(v_{i,k}^e,v_{m,2}^e) \le 2 \ne 3 \le d(y,v_{m,2}^e) \implies r(v_{i,k}^e \ | \ W) \ne r(y \ | \ W).
\end{align*}

It can be seen that every representation of vertices in $S$ are distinct. Therefore, $dim(G \ecp K_{m \times n}) = h(m(n-1)-2) + p$.
\end{proof}

An example of a resolving set in an edge comb product of a tree and $K_{3 \times 2}$ is given in Figure \ref{fig1}.

\begin{figure}[htbp]
\center{\includegraphics[scale=1]{Gambar/MD_ECP_Complete-Multipartites.png}}
 \caption{An edge comb product graph with metric dimension of $11$.} \label{fig1}
\end{figure}

Now, we will consider the edge comb product of arbitrary graph with Petersen graph $P(5,2)$.

\begin{theorem}
Let $T$ be a tree graph with size $h \ge 2$ and $p$ be the number of pendants in $T$. For any edge $e \in E(P(5,2))$, we have
\begin{align*}
    dim(T \ecp P(5,2)) = 2h.
\end{align*}
\end{theorem}

\begin{proof}
Let $e \in E(T)$ be an edge and $H^e$ be a subgraph of $G \ecp P(5,2)$ that is isomorphic to $P(5,2)$ and attached to an edge $e \in E(T)$. To prove that $dim(G \ecp P(5,2)) \ge 2h$, we will show that every subgraph $H^e$ contains at least two resolving vertices. We can split the problem into two cases.

\noindent \textit{Case 1.} Let $e$ be a pendant edge. Let $v_{1,2}^e$ be a vertex in $H^e$ which is adjacent to vertices outside $H^e$. It is clear that $H^e$ contains a resolving vertex. Assume that there exists $v_{i,j}^e \in V(H^e)$ as the only resolving vertex in $H^e$. This implies that $\{v_{1,2}^e,v_{i,j}^e\}$ is a resolving set for $P(5,2)$. This contradicts Theorem \ref{pet} which states $dim(P(5,2)) = 3$.

\noindent \textit{Case 2.} Let $e$ be a non-pendant edge. Let $v_{1,2}^e$ and $v_{2,2}^e$ be vertices in $H^e$ which are adjacent to vertices outside $H^e$. Assume that there exists $v_{i,j}^e \in V(H^e)$ as the only resolving vertex in $H^e$. Let $W$ be the resolving set of $G \ecp P(5,2)$. There exists vertices $x$ and $y$ in resolving set $W$ outside of $H^e$ such that $d(x,v_{1,2}^e) = a$ and $d(y,v_{2,2}^e) = b$. By considering only $W' = \{x,y\}$, we have the representation of vertices in $H^e$ below.
%\pagebreak
\begin{multicols}{2}
\noindent
    \begin{align*}
        r(v_{1,1}^e \ | \ W') & = (a+1,b+2),\\
        r(v_{2,1}^e \ | \ W') & = (a+2,b+1),\\
        r(v_{3,1}^e \ | \ W') & = (a+2,b+2),\\
        r(v_{4,1}^e \ | \ W') & = (a+2,b+2),\\
        r(v_{5,1}^e \ | \ W') & = (a+2,b+2),
    \end{align*}
%\columnbreak
%\noindent
    \begin{align*}
        r(v_{1,2}^e \ | \ W') & = (a,b+1),\\
        r(v_{2,2}^e \ | \ W') & = (a+1,b),\\
        r(v_{3,2}^e \ | \ W') & = (a+2,b+1),\\
        r(v_{4,2}^e \ | \ W') & = (a+2,b+2),\\
        r(v_{5,2}^e \ | \ W') & = (a+1,b+2).
    \end{align*}
\end{multicols}
It could be seen that $v_{3,1}^e, v_{4,1}^e, v_{5,1}^e$ and $v_{4,2}^e$ have the same representation with respect to $W'$. Since $diam(P(5,2)) = 2$, then
\begin{align*}
    \{d(v_{i,j}^e,v_{3,1}^e),d(v_{i,j}^e,v_{4,1}^e),d(v_{i,j}^e,v_{5,1}^e),d(v_{i,j}^e,v_{4,2}^e)\} \subseteq \{0,1,2\}
\end{align*}
This implies that there exists two vertices in $\{v_{3,1}^e, v_{4,1}^e, v_{5,1}^e, v_{4,2}^e\}$ which have the same representation with respect to $W$. This is a contradiction to the fact that $W$ is a resolving set.

By two preceding cases, it implies that every subgraph $H^e$ contains at least two resolving vertices. Consequently, $dim(G \ecp P(5,2)) \ge 2h$.

Next, we will show $dim(G \ecp P(5,2)) \le 2h$. Let $W$ be a resolving set of $G \ecp P(5,2)$. Let $e$ and $e'$ be edges in $E(T)$ which are not adjacent. Since for every edge $e \in E(T)$, $H^e$ contains two resolving vertices, then there exists $x \in W \cap H^e$ such that for every $v^e \in V(H^e)$ and $v^{e'} \in V(H^{e'})$ it holds
\begin{align*}
    d(x,v^e) \in \{0,1,2\} \qquad
    d(x,v^{e'}) \in \{2,3,4,5\}
\end{align*}
Likewise, there exists $y \in W \cap H^{e'}$ such that for every $v^e \in V(H^e)$ and $v^{e'} \in V(H^{e'})$ it holds
\begin{align*}
    d(y,v^{e'}) \in \{0,1,2\} \qquad
    d(y,v^e) \in \{2,3,4,5\}
\end{align*}
%%The only time where $v^e$ and $v^{e'}$ have same distances to $x$ and $y$ is when $d(x,v^e) = d(y,v^e) = d(x,v^{e'}) = d(y,v^{e'}) = 2$.
Let $v^e$ be a vertex with $d(x,v^e) = d(y,x^e) = 2$. If $y$ is a cut-vertex, then $d(x,v^e) < d(x,v^{e'})$. Else, if $y$ is not a cut-vertex, then $v^e$ must be cut-vertex. This implies $d(x,v^e) < d(x,v^{e'})$. In any case, there will be only one vertex in $V(H^e \cup H^{e'})$ which has $d(x,v^e) = d(y,x^e) = 2$. Consequently, for any resolving set $W$, every representation of $v^e$ and $v^{e'}$ will always be distinct.

Therefore, for a given set $W$, determining vertices in $H^e \cup H^{e'}$ in any two adjacent edges $e,e' \in E(T)$ have distinct representation with respect to $W$ is sufficient to check whether $W$ is a resolving set. Let $e$ and $e'$ be edges in $E(T)$ which are adjacent. We can consider a representation of each $H^e \cup H^{e'}$ with respect to a subset of $W$. This part can be split into three cases.

\noindent \textit{Case 1.} Let $e$ and $e'$ be pendant edges. Without loss of generality, let $v_{1,2}^e = v_{1,2}^{e'}$. %There exists a vertex $x$ in resolving set $W$ outside of $H^e$ such that $d(x,v_{1,2}^e) = a$.
By choosing $W' = \{v_{2,1}^e,v_{3,1}^e,v_{2,1}^{e'},v_{3,1}^{e'}\}$, we have
\begin{multicols}{2}
\noindent
    \begin{align*}
        r(v_{1,1}^e \ | \ W') & = (2,1,3,3),\\
        r(v_{2,1}^e \ | \ W') & = (0,2,4,4),\\
        r(v_{3,1}^e \ | \ W') & = (2,0,4,4),\\
        r(v_{4,1}^e \ | \ W') & = (1,2,4,4),\\
        r(v_{5,1}^e \ | \ W') & = (1,1,4,4),\\
        r(v_{1,2}^e \ | \ W') & = (2,2,2,2),\\
        r(v_{2,2}^e \ | \ W') & = (1,2,3,3),\\
        r(v_{3,2}^e \ | \ W') & = (2,1,4,4),\\
        r(v_{4,2}^e \ | \ W') & = (2,2,4,4),\\
        r(v_{5,2}^e \ | \ W') & = (2,2,3,3),
    \end{align*}
    \begin{align*}
        r(v_{1,1}^{e'} \ | \ W') & = (3,3,2,1),\\
        r(v_{2,1}^{e'} \ | \ W') & = (4,4,0,2),\\
        r(v_{3,1}^{e'} \ | \ W') & = (4,4,2,0),\\
        r(v_{4,1}^{e'} \ | \ W') & = (4,4,1,2),\\
        r(v_{5,1}^{e'} \ | \ W') & = (4,4,1,1),\\
        r(v_{2,2}^{e'} \ | \ W') & = (3,3,1,2),\\
        r(v_{3,2}^{e'} \ | \ W') & = (4,4,2,1),\\
        r(v_{4,2}^{e'} \ | \ W') & = (4,4,2,2),\\
        r(v_{5,2}^{e'} \ | \ W') & = (3,3,2,2).
\end{align*}
\end{multicols}

\noindent \textit{Case 2.} Let either one of $e$ or $e'$ be a pendant edge. Without loss of generality, let $e$ be a pendant edge and let $v_{1,2}^e = v_{1,2}^{e'}$. There exists a vertex $x$ in resolving set $W$ outside of $H^e \cup H^{e'}$ such that all shortest paths of $x$ and any vertices in $H^e \cup H^{e'}$ are containing $v_{2,2}^{e'}$. Let $a = d(x,v_{2,2}^{e'})$. Set $W' = \{v_{2,1}^e,v_{3,1}^e,v_{1,1}^{e'},v_{2,1}^{e'},x\}$. This implies
\begin{multicols}{2}
\noindent
    \begin{align*}
        r(v_{1,1}^e \ | \ W') & = (2,1,2,3,a+2),\\
        r(v_{2,1}^e \ | \ W') & = (0,2,3,4,a+3),\\
        r(v_{3,1}^e \ | \ W') & = (2,0,3,4,a+3),\\
        r(v_{4,1}^e \ | \ W') & = (1,2,3,4,a+3),\\
        r(v_{5,1}^e \ | \ W') & = (1,1,3,4,a+3),\\
        r(v_{1,2}^e \ | \ W') & = (2,2,1,2,a+1),\\
        r(v_{2,2}^e \ | \ W') & = (1,2,2,3,a+2),\\
        r(v_{3,2}^e \ | \ W') & = (2,1,3,4,a+3),\\
        r(v_{4,2}^e \ | \ W') & = (2,2,3,4,a+3),\\
        r(v_{5,2}^e \ | \ W') & = (2,2,2,3,a+2),
    \end{align*}
    \begin{align*}
        r(v_{1,1}^{e'} \ | \ W') & = (3,3,0,2,a+2),\\
        r(v_{2,1}^{e'} \ | \ W') & = (4,4,2,0,a+1),\\
        r(v_{3,1}^{e'} \ | \ W') & = (4,4,1,2,a+2),\\
        r(v_{4,1}^{e'} \ | \ W') & = (4,4,1,1,a+2),\\
        r(v_{5,1}^{e'} \ | \ W') & = (4,4,2,1,a+2),\\
        r(v_{2,2}^{e'} \ | \ W') & = (3,3,2,1,a),\\
        r(v_{3,2}^{e'} \ | \ W') & = (4,4,2,2,a+1),\\
        r(v_{4,2}^{e'} \ | \ W') & = (4,4,2,2,a+2),\\
        r(v_{5,2}^{e'} \ | \ W') & = (3,3,2,2,a+2).
\end{align*}
\end{multicols}
Figure \ref{fig2} illustrates the distinct representations of $H^e \cup H^{e'}$ for this case.

\begin{figure}[htbp]
\center{\includegraphics[scale=1.1]{Gambar/MD_ECP_Petersen_Sub.png}}
 \caption{Representations of two Petersen graphs $P(5,2)$ in a $T \ecp P(5,2)$.} \label{fig2}
\end{figure}

\noindent \textit{Case 3.} Let neither $e$ nor $e'$ be a pendant edge. Let $v_{1,2}^e = v_{1,2}^{e'}$. There exists a vertex $x$ in resolving set $W$ outside of $H^e \cup H^{e'}$ such that all shortest paths of $x$ and any vertices in $H^e \cup H^{e'}$ are containing $v_{2,2}^{e'}$. Similarly, there exists a vertex $y$ in resolving set $W$ outside of $H^e \cup H^{e'}$ such that all shortest paths of $x$ and any vertices in $H^e \cup H^{e'}$ are containing $v_{1,2}^e$. Let $a = d(x,v_{2,2}^{e'})$ and $b = d(y,v_{1,2}^e)$. By picking $W' = \{v_{1,1}^e,v_{2,1}^e,v_{1,1}^{e'},v_{2,1}^{e'},x,y\}$, it follows that
\begin{multicols}{2}
\noindent
    \begin{align*}
        r(v_{1,1}^e \ | \ W') & = (0,2,3,4,a+3,b+1),\\
        r(v_{2,1}^e \ | \ W') & = (2,0,2,3,a+2,b+2),\\
        r(v_{3,1}^e \ | \ W') & = (1,2,3,4,a+3,b+2),\\
        r(v_{4,1}^e \ | \ W') & = (1,1,3,4,a+3,b+2),\\
        r(v_{5,1}^e \ | \ W') & = (2,1,3,4,a+3,b+2),\\
        r(v_{1,2}^e \ | \ W') & = (1,2,2,3,a+2,b),\\
        r(v_{2,2}^e \ | \ W') & = (2,1,1,2,a+1,b+1),\\
        r(v_{3,2}^e \ | \ W') & = (2,2,2,3,a+2,b+2),\\
        r(v_{4,2}^e \ | \ W') & = (2,2,3,4,a+3,b+2),\\
        r(v_{5,2}^e \ | \ W') & = (2,2,3,4,a+3,b+1),
    \end{align*}
    \begin{align*}
        r(v_{1,1}^{e'} \ | \ W') & = (3,2,0,2,a+2,b+2),\\
        r(v_{2,1}^{e'} \ | \ W') & = (4,3,2,0,a+1,b+3),\\
        r(v_{3,1}^{e'} \ | \ W') & = (4,3,1,2,a+2,b+3),\\
        r(v_{4,1}^{e'} \ | \ W') & = (4,3,1,1,a+2,b+3),\\
        r(v_{5,1}^{e'} \ | \ W') & = (4,3,2,1,a+2,b+3),\\
        r(v_{2,2}^{e'} \ | \ W') & = (3,2,2,1,a,b+2),\\
        r(v_{3,2}^{e'} \ | \ W') & = (4,3,2,2,a+1,b+3),\\
        r(v_{4,2}^{e'} \ | \ W') & = (4,3,2,2,a+2,b+3),\\
        r(v_{5,2}^{e'} \ | \ W') & = (3,2,2,2,a+2,b+2).
\end{align*}
\end{multicols}

Let $W$ be the union of $W'$ from all $H^e \cup H^{e'}$ for every adjacent edges $e$ and $e'$. Since every vertices will have distinct representation, then $W$ is a resolving set with $|W| = 2h$. Hence, $dim(G \ecp P(5,2)) = 2h$.
\end{proof}

We present an example of a resolving set in an edge comb product of a tree and $P(5,2)$ in Figure \ref{fig3}.

\begin{figure}[thbp]
\center{\includegraphics[scale=1]{Gambar/MD_ECP_Petersen_Whole.png}}
 \caption{An edge comb product graph with metric dimension $8$.} \label{fig3}
\end{figure}

\section{Acknowledgement}
This research is supported by DIPA BOPTN UIN Syarif Hidayatullah Jakarta under contract B-207/LP2M-PUSLITPEN/TL.03/06/2021.

\pagebreak


\end{document}
